

**Round-1**

1. **Question:** The whole paper (especially the abstract) should be edited for english-language usage.

**Reply:** The abstract of the manuscript has been revised in English again.

2. **Question:** The aim of the study should be better explained in the introduction. Furthermore, a brief introduction on the uses of polidocanol in the control of bleeding should be added: Short-Term Results of Sclerotherapy with 3% Polidocanol Foam for Symptomatic Second- and Third-Degree Hemorrhoidal Disease. *J Invest Surg.* 2020 Apr 15:1-7. doi: 10.1080/08941939.2020.1745964 EndoTHEF: endo-luminal treatment of hemorrhoids with foam. *Ann Colorectal Res.* 2019;6(4):e86297. doi:10.5812/acr.86297 Safety and efficacy of foam sclerotherapy for treatment of low-flow vascular malformations in children. *J Vasc Surg Venous Lymphat Disord.* 2020 Nov;8(6):1074-1082

**Reply:** I have added a sclerosing agent to control bleeding in accordance with the editorial board's requirements, and have used the recommended articles.

3. **Question:** Did the authors follow-up the patient even after the month described? This is a limitation The authors should add all the limitations at the end of the manuscript.

**Reply:** After the patient received sclerotherapy, no complicati

ons were found during the telephone follow-up of one month.

After returning to the hospital for gastroscopy and endoscopic ultrasound, we found that the lesions disappeared.

4. **Question:** Discussion should be deeply shortened

**Reply:** We have reduced the discussion in the manuscript.

5. **Question:** I found the authors did not provide the original figures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor;

**Reply:** We used PowerPoint software to modify the pictures in the manuscript.

6. **Question:** I found the "Case Presentation" did not meet our requirements. Please re-write the "Case Presentation" section, and add "FINAL DIAGNOSIS", "TREATMENT", and "OUTCOME AND FOLLOW-UP" section to the main text, according to the Guidelines and Requirements for Manuscript Revision.

6 Re-Review: Required. 7 Recommendation: Conditionally accepted.

**Reply:** We have re-edited the "Case Presentation" as required by the magazine.

**Round-2**

**Question:** Congratulations on the changes made which have greatly improved the manuscript. However, there are some points to clarify: - The whole paper (and not just the abstract) should be edited for english-language usage. Indeed, there are numerous grammatical errors and linguistic inaccuracies. Especially in the introduction. - The authors reported the ref. 7 in the text as "Pierluigi". Unfortunately it is not correct. The best way is "According to Lobascio et al." or other ways. Therefore it is necessary to modify both the sentence and to change from name (Pierluigi) to surname (Lobascio)

**Reply:** The English grammar of the full text of the manuscript has been revised again according to the feedback from the editorial office